Author's response to reviews

Title: Up-regulated expression of type II very low density lipoprotein correlates with cancer metastasis and has a potential link to beta-catenin in different cancers

Authors:

Lei He (atoprine@qq.com)
Yanjun Lu (junyanlu_2000@163.com)
Peng Wang (wpcz_1003@163.com)
Jun Zhang (zhangjun1008@126.com)
Chuanchang Yin (yinchuanchang8260464@126.com)
Shen Qu (qsc_n@sina.cn)

Version: 3 Date: 2 August 2010

Author's response to reviews: see over
Dear editors:

Here within enclosed is a manuscript by Lei He, Yanjun Lu, Peng Wang, Jun Zhang, Chuanchang Yin and Shen Qu titled “Up-regulated expression of type II very low density lipoprotein correlates with cancer metastasis and has a potential link to β-catenin in different cancers”. It is submitted to be considered for publication as a “Research article” in your journal. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.

We believe the paper may be of particular interest to the readers of your journal because the study it reports stated the higher very low density lipoprotein II (VLDLR II) expression levels are associated with lymph node and distant metastasis in different human cancers, and there are highly significant positive correlations between VLDLR II and β-catenin in gastric and breast cancer tissues. To our knowledge, This is the first to reveal the closer relationship of VLDLR II with clinicopathological information. This study will help to thoroughly understand the function of VLDLR II and to open new perspectives for diagnosis and treatment of a wide range of cancers.

Correspondence and phone calls about the paper should be directed to Shen Qu at the following address, phone and fax number, and e-mail address:

Shen Qu, Department of Biochemistry and Molecular Biology, Tongji Medical College, Huazhong University of Science and Technology, Wuhan 430030, China.
Phone: +86-27-83662630; Fax: +86-27-83662645; e-mail: qsc_n@sina.cn

Thanks very much for your attention to our paper.

Sincerely yours

Shen Qu

August 1, 2010